%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\documentstyle[12pt]{article}
\topmargin-3cm
\textheight 26cm
\textwidth17.5cm
\oddsidemargin-1cm
\evensidemargin-1cm
\pagestyle{empty}
\begin{document}
{\large\bf Math\,425, section 1}\hfill {\large\bf Fall 1997}

\begin{center} {\Large\bf Complex Variable Theory }
  \\[3ex]
{\bf Classes meet MWF 3:00 -- 3:50 p.m. in Room 311 DeBartolo}
\end{center}

\mbox{}\hspace*{-.2em}\begin{tabular}{lp{12.7cm}}
{\bf Instructor:}& Heide Gluesing-Luerssen,
                   CCMB 300-A, phone 631-8698\\
                 & e.mail: Heide.Gluesing-Lueerssen.1@nd.edu\\[1.3ex]
{\bf Office hours:} & Monday 10:40 -- 12:00,
                      Thursday 1:00 -- 2:30 p.m.,
                      and by Appointment\\[1.3ex]
{\bf Text:} & Stephen D. Fisher: Complex Variables, 2nd Edition,
              Wadsworth \& Brooks/Cole, 1990\\[1.3ex]
{\bf Homework:} & Homework will be assigned (almost) every class.
                  There will be 5 marked exercises each week, which are due
the
                  following Monday, they will be graded.
                  Late Homework will not be graded.\\
                & The other exercises are suggestions, which you should take
                  serious as well.
                  As you know from expirience, doing a lot of exercises will
                  give you more practice and a better understanding of the
                  material.\\[1.3ex]
{\bf Exams:} & There will be three Midterm Exams and a two-hour Final on the
               following dates:\\[.5ex]
             &  \mbox{}\qquad\begin{tabular}{lll}
                  Exam 1\qquad\ & Friday, Sept.~19,& 3:00 p.m. in class\\
                  Exam 2 & Friday, Oct.~17, & 3:00 p.m. in class\\
                  Exam 3 & Friday, Nov.~21, & 3:00 p.m. in class\\
                  Final      & Wednesday, Dec.~17, & 4:15 -- 6:15 p.m.
                \end{tabular}\\[.3ex]
             & All exams and the final will take place in Room 311 DeBartolo.
               \\[.5ex]
             & Exams may be made up only with an excused absense from the
               Associate
               Vice President for Residential Life.
               Conflicts with exams in other courses must be resolved
during the
               first week of classes.\\[1.3ex]
{\bf Quizzes:} & There may also be some quizzes as need arises.\\[1.3ex]
{\bf Honor Code:} & {\em You are reminded that you are bound to the Notre
Dame Honor
                Code.} \\[.5ex]
               & You may coorporate with other classmates and consult books
and
                 notes
                 while working on your homework, but
                {\em do not copy each other\/}!\\
               & During exams and quizzes the use of books, notes,
                 etc.\ and
                 communication with others is not allowed.\\[1.3ex]
{\bf Grading:} & \mbox{}\hspace*{-.3em}\begin{tabular}[t]{lr}
                      Exam 1 -- 3:\quad 3 $\times$ 100 =& 300 \\
                      Final:                           & 150 \\
                      Homework \& Quizzes              &  50 \\ \hline
                      {\bf Total Score}:               & {\bf 500}
                 \end{tabular}\\[10ex]
{\bf Class}& It is very important that you miss no class at all,
              and take good notes\\
{\bf Attendance:}& on every lecture.
              Then before the next lecture read your class notes and the
book, and do
              the homework based on the material just covered.
              Try to have your questions answered before the next class begins.
              You are strongly encouraged to form small groups of study with
              your classmates to discuss the material and also the Homework
              problems.\\[1.3ex]
{\bf Note:} & You are encouraged to use my office hours and contact me early
              with any difficulties, before problems accumulate beyond help.
\end{tabular}
\end{document}
